Author's response to reviews

Title: Vulnerability of eco-environmental health to climate change: The views of government stakeholders and other specialists in Queensland, Australia

Authors:

Linn B Strand (l.strand@qut.edu.au)
Shilu Tong (s.tong@qut.edu.au)
Rosemary Aird (r.aird@qut.edu.au)
David McRae (david.mcrae@climatechange.qld.gov.au)

Version: 3 Date: 23 July 2010

Author's response to reviews:

Dear editor,

We have formatted the manuscript as requested and have:

- Referred to table 2 in the text.
- Removed all highlighting and tracking form the manuscript
- Provided up to 30 author names for each citation. The term et al. is not used if there are fewer than 30 authors.
- Removed table placement text from the main text
- Checked the manuscript for any typographical errors and made a few final corrections

Thank you for accepting our manuscript!

Kind Regards,

Linn Strand, Shilu Tong, Rosemary Aird and David McRae